Author's response to reviews

Title: A qualitative study of perceived risk for HIV transmission among police officers in Dar es Salaam, Tanzania

Authors:

Edith A.M Tarimo (etarimo54@yahoo.co.uk)
Thecla W. Kohi (thecla.kohi@gmail.com)
Muhammad Bakari (drbakari@yahoo.com)
Asli Kulane (Asli.Kulane@ki.se)

Version: 4 Date: 30 July 2013

Author's response to reviews: see over
Author’s response to reviews

Title: A qualitative study of perceived risk for HIV transmission among police officers in Dar es Salaam, Tanzania

Authors:

Edith A.M Tarimo (etarimo54@yahoo.co.uk)
Thecla W. Kohi (thecla.kohi@gmail.com)
Muhammad Bakari (drbakari@yahoo.com)
Asli Kulane (Asli.Kulane@ki.se)

Authors’ response to reviews: see over
The BMC Public Health Editorial Team

Object: **MS: 1771776190842824** - A qualitative study of perceived risk for HIV transmission among police officers in Dar es Salaam, Tanzania

Thank you for consideration of our manuscript for publication in your journal. We have reviewed the above manuscript according to your suggestions.

**Quality of written English:** I have sought further professional assistance from a native English speaking scientific copyeditor namely Karen Schmidt (See the attached draft with tracks).

Sincerely Yours

[Signature]

Edith A.M Tarimo